vc/6-Oct-03
Week 41 starting 06 Oct 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

 Today(07/10//03)
	Forward look,   & Next Magnet 

	TBA1
	3047
	Should be removed
	Bench to be de-activated – preparation for D bench

	TBA2
	3042
	Should be removed
	 Next Magnet 1020

	TBC1
	3045
	On Standby
	Start Cold Tests on Fri &  during w’end when SSS not used

	TBC2
	SSS1
	cold
	Cold tests

	TBE1
	2002
	Warm HV Tests
	Disconnect Wed. & prepare for New without Anticryostats-3044?

	TBE2
	3039
	cold
	Cold Tests till Thurs?

	TBF1
	2025 after F2 done? 
	
	Prepare For 

Commissioning AFTER F2

	TBF2
	2025
	Cold
	Commissioning TESTS


Test Coordinator this week:  Marco+Nikolai;  MTM  Tech. on call:  (16-3602) : Maryline

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. New RRR Doc.;  

2. F Bench Commissioning & Red folder 

3.   “  “”           “”                 ( console from CO ?).,

4. Interactions in Traveller & Info 

5. New Test Sequence Update Web form & Display(GHH)

6. AOB

· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 30_M


	TBA2 : No data from quench (MIITS), 
	
	AQA
	?

	Tuesday 30_A


	TBE1:IAP @warm failed.: D20 connector not connected
	MTM
	
	Connected and OK

	Tuesday 30_A


	TBA2: No read current during special LHC cycle between 760 A and 3 kA
	MTM
	Display
	Read current in black during the ramp for LHC cycles.

	Wednesday 1_A


	TBE2: IAP @warm forgotten. Config file not ready.
	MTM
	
	Prep 7 has not to be performed before prep6 . Wait for the config file to perform prep 6.

	Wednesday 1_A


	Cryogenic problem (warmer not working)
	MTM
	
	Magnets lost the cryo OK and go to 4.4 K 

	Wednesday 1_N


	MM (TBA2) : 10 kA-LHC cycle launched in Qloc position. Over ranges observed.
	MTM
	
	Test aborted and repeated.

	Wednesday 1_N

Thursday 2_M


	MM: After export data for MM, display is hung
	AB/CO
	Display
	Pedro informed.

Investigation on progress

	Thursday 2_M


	TBA1: Unable to launch “thermal cycle “ from TEMA
	AB/CO
	
	One INCAA card was removed and MXI crate powered off. Solved by Magnus.

	Thursday 2_N

Friday_3_M


	TBA1: HV test at 1.9 K failed: problem with humidity in current leads
	MTM
	
	Test MB/QH and QH/ground.

OK

Test coordinator gave oK to continue the test at cold.

But before warm up complete HV test (current leads disconnected) performed.

	Thursday 2_N


	TBA1: IAP at cold . MXI error


	MTM
	
	HF data acquisition card missing

	Thursday 2_N


	TBA2 : Quench at 6.5 kA to warm up

LF triggered but no File not created
	MTM/AB/CO?
	LF
	

	?Thursday 2_N


	Cryogenic : cut of the power in the Liquefactor. Then Temperature not stable.


	MTM
	Liquefactor
	NO HV test possible,

	Friday 3_A


	TBE1: IAP @ cold. Problem to launch the test,
	MTM
	IAP
	Switch off and on the power supply.

	Friday 3_N


	TBA1: Problem in Power Converter. 


	SL/PO
	
	Loops were changed in RMA,RME, RMC without informing us. Loop of SSS. Giorgio changed the loop. 

	Saturday 4_M


	TBE1: HV test at cold failed 

But data could not be found


	AB/CO?
	
	Test repeated.

	Saturday 4_M


	TBE1: HV test at cold failed 

Ext Passive bus bar.


	MTM
	
	Giorgio and Stephane informed. 

OK to continue the test.

Non conformity was opened.

	Saturday 4_M


	TBE1:D_2_U-L potaim card did not work


	MTM
	
	Giorgio informed.

Replaced by one in the cluster F

	Saturday 4_A


	TBA2:IAP@warm failed

MXI communication error for HF acqusition


	MTM-AB/CO?
	IAP
	Acquisition no getting armed

HF and LF acquisition rebooted.

	Sunday 5_M


	TBA1:RRR measurement DVM DAQ system blocked 


	AB/CO
	DVM DAQ
	DVM DAQ rebooted.

	Sunday 5_N


	Electrical power cutStop of the liquefactor  . Cryo not OK for all benches


	Liquefactor
	
	


